Author's response to reviews

Title: Vitamin A deficiency and inflammatory markers among preschool children in the Republic of the Marshall Islands

Authors:

Maria Maqsood (maria_maqsood@yahoo.com)
Barbara Dancheck (bdanche1@jhmi.edu)
Mary V Gamble (mvg7@columbia.edu)
Neal A Palafox (npalafox@hawaii.edu)
Michelle O Ricks (mricks2@jhmi.edu)
Kennar Briand (npalafox@hawaii.edu)
Richard D Semba (rdsemba@jhmi.edu)

Version: 2 Date: 22 November 2004

Author's response to reviews:

Dear Dr. Hiromichi Kumagai:

We have revised the manuscript as follows:

The reviewer 2 asked that we modify the abstract to include a sentence in the conclusions that the exclusion of children with elevated acute phase proteins does not improve prevalence estimates. We have modified the conclusion to include this phrase.

Thank you.

Sincerely,

Richard Semba MD MPH
Associate Professor
Johns Hopkins School of Medicine